Author's response to reviews

Title: Reducing Depression in Older Home Care Clients: Design of a Prospective Study of a Nurse-Led Interprofessional Mental Health Promotion Intervention

Authors:

Maureen F Markle-Reid (mreid@mcmaster.ca)
Carrie McAiney (mcaineyc@mcmaster.ca)
Dorothy Forbes (dorothy.forbes@ualberta.ca)
Lehana Thabane (thabanl@mcmaster.ca)
Maggie Gibson (Maggie.Gibson@sjhc.london.on.ca)
Jeffrey S Hoch (jeffrey.hoch@utoronto.ca)
Gina Browne (browneg@mcmaster.ca)
Thomas Peirce (tom.peirce@hnhb.ccac-ont.ca)
Barbara Busing (barbara.busing@hnhb.ccac-ont.ca)

Version: 3 Date: 2 August 2011

Author's response to reviews:

August 2, 2011

Rachel Neilan, MSc
Executive Editor
BMC Geriatrics

Dear Ms. Neilan:

RE: MS: 4046258775487766

Reducing Depression in Older Home Care Clients: Design of a Prospective Study of a Nurse-Led Interprofessional Mental Health Promotion Intervention

Thank you for your letter inviting us to revise and resubmit our manuscript to BMC Geriatrics. We have registered our trial and included the trial registration number in the last line of the abstract of the manuscript both online and in the main manuscript file.

We trust that we have adequately addressed your comments. We look forward to your response to this revised version and the opportunity of publishing this study protocol in BMC Geriatrics.

Yours sincerely,
Dr. Maureen Markle-Reid, RN, MScN, Ph.D.
Acting Assistant Dean (Research) and Associate Professor, School of Nursing
Associate Member, Department of Clinical Epidemiology and Biostatistics
McMaster University
1200 Main Street West, HSC 3N28H
Hamilton, Ontario, Canada L8N 3Z5
Tel: (905) 525-9140, ext. 22306; Fax: (905) 521-8834
E-mail: mreid@mcmaster.ca